



Luki Trihardani ¹, Chi-Tai Wang ^{1,*} and Ying-Jiun Hsieh ²

- ¹ Institute of Industrial Management, National Central University, Taoyuan 320, Taiwan; luki.trihardani@uisi.ac.id
- ² Graduate Institute of Technology Management, National Chung Hsing University, Taichung 402, Taiwan; arborfish@dragon.nchu.edu.tw
- * Correspondence: ctwang@mgt.ncu.edu.tw; Tel.: +886-3-427-5033

Abstract: The adoption of renewable energy (RE) is a promising business strategy for bed and breakfasts (B&Bs) to mitigate climate change while maintaining a competitive edge. However, there is still a lack of analytical studies to determine an optimal RE mix for tourism accommodations. This study thus proposes a practical approach to enable all B&Bs to make optimal RE decisions for their facility. A mixed-integer programming (MIP) model is developed and tested in a case study. The model successfully identifies an optimal hybrid energy system for two scenarios, the base case that generates 116,942 kWh of electricity annually at the cost of USD 21,499, and the unconventional technology case that generates 114,474 kWh of electricity annually at the cost of USD 24,670. Compared to purchasing all the required electricity from the power grid, both scenarios can save more than 26 tons of CO2e/year. The analysis provides valuable information for B&Bs to initiate a smooth energy transition with affordable costs. This study considers various energy components, including hybrid RE, batteries, the power grid, self-sufficiency targets, and various RE technologies. Therefore, B&Bs can choose a preferred self-sufficiency target where RE satisfies a specific portion of the energy demands and the power grid satisfies the rest. The model can also evaluate the tradeoff between investing in RE technologies and purchasing larger batteries. These findings will assist B&Bs in accelerating the adoption of RE globally.

Keywords: bed and breakfast; renewable energy; tourism accommodation; emission; optimization

1. Introduction

Climate change has brought various threats to humanity. Among them, global warming is arguably the most serious. According to the World Meteorological Organization, 2011–2020 was the warmest decade since the 1800s [1]. Global warming comes from humaninduced emissions, mainly from burning fossil fuels [2]. As a significant contributor to the world economy, international tourism's revenue was USD 1481 billion in 2019, a 4.8% year-on-year increase [3]. However, tourism emissions increased from 3.9 to 4.5 GtCO₂e between 2009 and 2013, representing 8% of global emissions [4]. Tourism can only be sustainable in the long term if it transforms into a climate-resilient industry [5]. Therefore, there is an urgent need for the tourism industry to embrace a low-carbon pathway to mitigate global warming and ultimately achieve sustainable tourism. However, sustainable tourism is a continuous process that requires seamless collaboration among tourism stakeholders, including the business, local community, government, tourists, and international community [6]. Instead of focusing on only one aspect (economic growth), sustainable tourism considers a balance between economic, environmental, and social-cultural aspects to ensure the long-term sustainability of tourism development [7]. These considerations lead the tourism industry to achieve productivity while preserving the natural environment and maintaining the quality of life. However, under a business-as-usual scenario,



Citation: Trihardani, L.; Wang, C.-T.; Hsieh, Y.-J. Making Optimal Location-Sizing Decisions for Deploying Hybrid Renewable Energy at B&Bs. *Appl. Sci.* **2022**, *12*, 6087. https://doi.org/10.3390/ app12126087

Academic Editor: Matti Lehtonen

Received: 9 April 2022 Accepted: 13 June 2022 Published: 15 June 2022

Publisher's Note: MDPI stays neutral with regard to jurisdictional claims in published maps and institutional affiliations.



Copyright: © 2022 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). tourism emissions are projected to increase by 169% between 2010 and 2050 [8]. Since reducing emissions is a challenge for sustainable development, this finding recognizes that sustainability practices in the tourism industry are still insufficient [5].

Sustainable tourism should optimize environmental resources, a vital element of tourism development [9]. However, accommodation is a resource-intensive tourism subsector [10,11]. As more than 70% of energy demand is consumed for electricity [10], accommodation is the second-highest emitter after transportation, accounting for 21% of tourism emissions [12]. This study focuses on a particular type of accommodation, bed and breakfast (B&B), which has experienced rapid growth through sharing-economy platforms [13]. As a small tourism business owned and operated by a family, a B&B is an innovative sideline business built and developed from a family's underutilized assets [14]. Unfortunately, many B&Bs have recently experienced a significant decline in guest numbers. For example, in Taiwan [15] and Northern Ireland [16], the occupancy rate was below 25% from 2013 to 2018. Oversupply and homogeneity are the reasons for this severe challenge [17,18]. Therefore, B&Bs need to build a competitive advantage in a saturated market by differentiating themselves from the competition with novel features. Recent literature has investigated whether environmentally friendly practices in hotels (e.g., waste management, energy efficiency) can increase customer satisfaction, a critical factor in hotel selection [19]. In fact, the adoption of renewable energy (RE) as a primary energy source is the least adopted by accommodations [20]. However, B&Bs can improve energy security by changing their energy sources while reducing energy costs with a short payback period, which has been implemented in real-world applications [21]. Using low-carbon energy in tourism destinations also contributes to community economies by generating revenue, creating jobs, and diversifying income [22]. Therefore, energy transition would not only reshape a B&B's image as an eco-friendly accommodation business [23] but also facilitate the sustainable development of local communities.

In addition, the shift to RE has the potential to maintain the B&B's competitive advantage by offering new service innovations continuously. Considering that visitors are more likely to stay at B&Bs than traditional hotels due to their authenticity [24,25] and social interaction [26], B&Bs can collaborate with the entire community to offer novel experiences that leverage and integrate local resources [27]. Another interesting innovation is the increasing adoption of electric vehicles (EVs), which is expected to increase six-fold between 2019 and 2030 [28]. As pioneered by several hotel chains [29,30], B&Bs can offer EV charging services at their facilities to proactively respond to market changes, generate new revenue streams, and develop more innovations. In terms of the internal conflict due to energy transition, B&Bs can take the opportunity of transgenerational ownership (succession from the old to the young generation). When the new management reevaluate their business strategy during this period [14,31], this would be an ideal time to conduct an energy transition. A significant reduction in RE costs, even below 3.0 US cents/kWh in some countries [32], means it is also an ideal time for initiating the transition. Unfortunately, despite these opportunities, RE is still underutilized in the accommodation industry. For example, only 17.3% of hotels in the Asia-Pacific region have installed renewable energy generation devices (REGDs) [33]. The most common reason for not deploying RE at the accommodations is the personal belief of (high) upfront investment [34]. Compared to other accommodations, the lack of capital reserves [35] and heavy dependence on the power grid in most B&Bs [36] pose a challenge to a successful energy transition. Furthermore, there is a lack of know-how in initiating energy transition in accommodation businesses [34], mainly due to several complexities. B&Bs have to balance fluctuations in energy production and consumption due to the intermittent nature of most RE energy sources and the occurrence of peak demand at certain times.

The hybrid RE system that combines two or more energy sources (e.g., solar and wind) can adequately manage these fluctuations [37]. However, a hybrid RE system, often equipped with batteries, increases the complexity of the problem compared to a single-source RE system. Further complexity arises from the variety of REGDs on the market,

including conventional REGDs (e.g., solar panels and wind turbines) and unconventional REGDs (e.g., the Wind Tree). Both offer different tradeoffs that B&Bs must evaluate when transitioning to RE. B&Bs also need to manage the site availability since an installed RE system requires more land than a fossil power plant [38]. There is an urgent need for a decision-making framework to manage this complexity to ensure a resilient energy supply at an affordable cost. The decision should include the type and size of RE technologies for each suitable location on the B&B. According to the systematic review, the optimization approach provides good energy planning for ensuring sustainability [39]. Therefore, this study proposes a practical approach to address the critical need for all B&B businesses to make optimal decisions on deploying hybrid RE technologies in their locations. This study provides a model that integrates system components, including hybrid RE systems, batteries, the power grid, different ratios of self-sufficiency, and different RE technologies, to ensure a smooth energy transition. This integration provides B&Bs the flexibility to choose their preferred self-sufficiency as a target to be achieved by allocating a specific share of the energy demand to be met by RE and the rest by the power grid. In addition to bridging the gap between the high cost of RE technologies and the affordability of B&Bs, the model can also evaluate the tradeoff between conflicting goals, costs, and emissions when designing the system. This integration also assists the B&B in evaluating the tradeoff between conventional and/or unconventional REGDs and investigating the need to determine a specific size of batteries to regulate energy flow. The mixed-integer programming (MIP) model is developed as a proposed method to solve the problem. The model then demonstrates its capabilities in a case study on a simulated B&B facility.

The rest of this paper is organized as follows. Section 2 reviews related research. Section 3 presents the optimization model. Section 4 describes the methodology, followed by the results and discussion in Section 5. Finally, Section 6 concludes this research with suggestions for future studies.

2. Literature Review

There is limited research investigating the feasibility of using RE in accommodation businesses [40]. Among those papers, design system optimization, that is, identifying a reliable and affordable system configuration, has been the main focus. Due to highly complicated dynamics between the major factors (costs, system performance, and energy demand/supply management), designing such an RE system is challenging. Therefore, a cost-effective energy solution requires an appropriate combination of RE systems [41]. Generally speaking, simulation-based optimization is arguably the most popular tool for solving this kind of problem [42]. Hybrid optimization of multiple energy resources (HOMER) [43] is one of the most well-known software developed for such purposes, with cost minimization as the primary objective. At a high level, HOMER works by simulating various system configurations based on user inputs [44]. Limited by its optimization technique, HOMER does not always produce an optimal solution. To be precise, when there are a large number of system configurations, HOMER often identifies only a suboptimal solution. Some researchers [45,46] also use mathematical optimization models, and the results are more cost-effective than HOMER.

Despite these issues, several tourism-related papers have used HOMER to evaluate the usage of RE at accommodation facilities. Dalton et al. [47] conducted a cost study and evaluated the allocation of various configurations, including solar or wind energy (i.e., stand-alone RE), diesel-only, and diesel–RE. Using HOMER, these simulations revealed the feasibility of using renewables to meet all energy needs in an Australian resort. Another finding was that diesel–RE would cost 30% less than the other configurations. Furthermore, Dalton et al. [48] extended the research to investigate the current and future feasibility of a stand-alone hybrid solar–wind system using three tourist hotels as case studies. Güler et al. [49] used HOMER to determine the optimal size of RE technologies, including solar panels, wind turbines, batteries, and converters, for an uninterrupted power supply for a Turkish hotel. Since the hotel had a surplus RE (that is, RE neither consumed by the hotel nor saved in the battery) in certain time periods, it would be able to open up new revenue streams for the hotel. Fazelpour et al. [50] investigated the environmental benefits of deploying an RE system at an Iranian hotel with HOMER. When emission costs were considered, the most effective configuration changed from a diesel-only to a dieselwind system. Using HOMER, Shezan et al. [51] analyzed the performance of a stand-alone diesel-wind system to power an eco-lodge in the highlands of Malaysia. In addition to the applicability of the RE system for supplying remote areas, this study also proves that the optimal allocation of renewables requires accurate weather data. Khan et al. [52] explored the competitiveness of various hybrid systems with HOMER for supplying energy in an island resort. A diesel-solar-wind-hydro system had the highest RE penetration compared to others. However, this configuration is not feasible since only a certain amount of land is available for solar panel installation. Finally, Hossain et al. [53] demonstrated that a hybrid energy system consisting of two or more RE sources used together (e.g., dieselsolar–wind system) is more viable than energy systems with only one type of energy source. Another finding was that energy demand may vary greatly depending on the season due to occupancy rates, which was also studied by Borowski et al. [54].

Another simulation-based tool called transient system simulation program (TRN-SYS) [55] has also been used to study energy problems for tourism facilities. This software offers flexibility by allowing users to modify their models [56] or adjust the component library when necessary [57]. Buonomano et al. [58] used TRNSYS to simulate a geothermalsolar energy system at an Italian resort. This study proves the importance of oil tanks used as energy storage to regulate the fluctuating solar energy supply. Beccali et al. [59] confirmed the critical role of a battery in balancing the intermittency of renewables. Using TRNSYS, their results highlight that the battery can reduce the purchased electricity from the power grid by up to 61%. In addition to simulation, heuristic algorithms are also a popular tool for solving RE problems in accommodation facilities. Soheyli et al. [60] developed a multi-objective particle swarm optimization (PSO) algorithm to minimize costs, land requirements, and CO₂ emissions for an Iranian hotel. According to their results, an integrated RE system for cooling, heating, and power generation is less costly than a separate system. Meschede et al. [61] showed that occupancy fluctuations between weekends and weekdays can significantly affect energy demand. Therefore, generating synthetic time series data of energy demand based on simulated occupancy and weather data would be useful in providing a realistic setting. The levelized cost of electricity (LCOE) is also used to evaluate RE configurations from a financial perspective as it is able to capture long-term economic sustainability.

Table 1 presents the key features of tourism accommodation papers on the system optimization of RE systems. It also includes the cost of energy for comparing the average cost per kWh for energy generation. From this table, findings and proposed contributions can be summarized as follows.

- Most prior studies do not determine the optimal location for deploying REGDs. Although Soheyli et al. [60] investigate the total area required, they do not investigate which locations are suitable for each RE technology. This study proposes the optimal location-sizing decisions on deploying a hybrid RE system at the B&B location to fill this gap. To the author's knowledge, this is the first study in tourism accommodations that determines the optimal size of the RE system for each suitable location.
- While simulation often provides suboptimal solutions, all of the prior studies use simulation-based optimization using software [47–53,58,59] and heuristic algorithms (PSO) [60,61]. Therefore, the MIP model is developed in this study as a proposed method to solve the problem.
- Since the cost of implementing 100% RE is relatively high for some B&Bs, integrating the RE system with the power grid would be an attempt to bridge the high costs of RE and the affordability for B&Bs. Based on the literature, only the research of [49,61] include the power grid in the RE system. This integration also offers more flexibility in evaluating the conflicting objectives of energy planning (costs vs. emissions).

Interestingly, almost half of the prior studies do not consider the volume of emissions that can be reduced in the RE system [48,49,58,59,61]. This study attempts to fill this gap by integrating the power grid into the RE system to eventually reach 100% RE. Therefore, the model would allocate a specific portion of energy demands that RE must meet and the power grid supplies the rest (e.g., 50% RE, 60% RE, ..., 100% RE).

- All prior studies use a specific type of solar panel and/or wind turbine for the RE system. They neglect the existence of various REGDs on the market, especially newgeneration RE technologies, which may provide higher energy generation at higher costs. This study evaluates various types of solar and wind REGDs, both conventional and unconventional, to assist B&Bs in investing in RE technologies.
- With regards to energy demand, only Meschede et al. [61] consider that higher occupancy on weekends creates a higher energy demand than on weekdays. This study uses this finding when establishing energy demand data for the B&B.

In summary, various energy components are integrated into a model to ensure smooth energy transition, including hybrid RE, batteries, the power grid, different self-sufficiency targets, and different RE technologies.

	Accommodation	Optimization	Cost of	System Component					F · · 6	Opti	mization	PDW ⁷
Prior Study	Types (No. of Room)	Software Used	(USD/kWh)	SP ¹	WT ²	OT ³	BT ⁴	GR ⁵	Emission °	Size	Location	
Dalton et al. [47]	Resort (378)	HOMER	n/a	\checkmark	\checkmark	х	\checkmark	х	\checkmark	\checkmark	х	х
Dalton et al. [48]	Resort (34)	HOMER	n/a	\checkmark	\checkmark	х	\checkmark	х	х	\checkmark	х	х
Güler et al. [49]	Hotel (133)	HOMER	1.02	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark	х	\checkmark	х	х
Fazelpour et al. [50]	Hotel (125)	HOMER	0.34	\checkmark	\checkmark	x	\checkmark	x	\checkmark	\checkmark	x	x
Shezan et al. [51]	Lodge (100)	HOMER	0.27	х	\checkmark	х	х	х	\checkmark	\checkmark	х	х
Khan et al. [52]	Resort (n/a)	HOMER	0.14	\checkmark	\checkmark	х	\checkmark	х	\checkmark	\checkmark	х	х
Hossain et al. [53]	Resort (268)	HOMER	0.28	\checkmark	\checkmark	х	\checkmark	х	\checkmark	\checkmark	х	х
Buonomano et al. [58]	Resort (17)	TRNSYS	0.21	\checkmark	x	GW ⁸	\checkmark	x	х	\checkmark	x	x
Beccali et al. [59]	Hotel (20)	TRNSYS	n/a	\checkmark	х	STC 9	\checkmark	х	х	\checkmark	х	х
Soheyli et al. [60]	Hotel (n/a)	PSO	n/a	\checkmark	\checkmark	х	\checkmark	х	\checkmark	\checkmark	\checkmark	х
Meschede et al. [61]	Hotel (n/a)	PSO	0.29	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark	x	\checkmark	x	\checkmark

Table 1. Key features of selective tourism accommodation papers.

¹ SP = solar panels; ² WT = wind turbines; ³ OT = other REGDs; ⁴ BAT = batteries; ⁵ GR = the power grid; ⁶ Emission = emissions consideration; ⁷ PDW = peak demand on weekend; ⁸ GW = geothermal wells; ⁹ STC = solar thermal collectors; n/a = not available in the paper.

3. Optimization Model

An MIP model is presented in the following section to serve as the main decisionmaking tool. By considering all available locations at the B&B suitable for RE generation, the proposed model will select appropriate REGDs to operate at each location, generating sufficient RE to meet the B&B's energy demands. If necessary, this model will also use batteries (with a suitable storage capacity) to facilitate RE usage. As most B&Bs are a small tourism business run by a single family, they certainly want to make every investment dollar count. Therefore, the model's objective is set to identify a particular deployment of REGDs so that the required investments are as low as possible. The model's details are as follows.

Use the rooftop as an example. Suppose a particular B&B can choose solar panels, small wind turbines, or a Tesla solar roof [62] to generate RE from their rooftop. In this case, the rooftop is then an element of L and when l is indicating the rooftop, DVC_l is then a set containing three elements: solar panels, small wind turbines, and a Tesla solar roof.

For example, if *d* represents a solar panel to be installed at the rooftop, UNT_d is approximately 1.67 m².

Continuing from the above example, if *d* is a solar panel and *l* is the rooftop, $Q_{d,l}$ will indicate the total number of solar panels to be installed at the B&B's rooftop.

 $E_{l,t}$ is the total energy (kWh) generated by those REGDs deployed at location *l* during planning period *t*.

B is the required number of batteries.

 I_t is the RE that is saved in the batteries at the end of the planning period t (kWh). W_t is the RE that is no use to the B&B in the planning period t (kWh). The following is the proposed MIP model

minimize
$$B * CBT + \sum_{t \in T} \sum_{l \in L} \sum_{d \in DVC_l} Q_{d,l} * ENR_{d,t} * CST_d$$
 (1)

Subject to

$$E_{l,t} = \sum_{d \in \text{DVC}_l} Q_{d,l} * \text{ENR}_{d,t} \quad \forall l \in \text{L}, \ t \in \text{T}$$
(2)

$$I_t = \sum_{l \in \mathbf{L}} E_{l,t} - \operatorname{REQ}_t - W_t \ t = 1$$
(3)

$$I_t = I_{t-1} + \sum_{l \in \mathcal{L}} E_{l,t} - \operatorname{REQ}_t - W_t \quad \forall t \ge 2$$
(4)

$$I_t \le B * BCL \qquad \forall t \in T$$
 (5)

$$\sum_{d \in \text{DVC}_l} Q_{d,l} * \text{UNT}_d \le \text{ARA}_l \quad \forall l \in \mathcal{L}$$
(6)

$$B, I_t, W_t, E_{l,t} \ge 0, \quad Q_{d,l} \ge 0 \text{ and integer}$$
(7)

The model's objective, Equation (1), is to minimize the total costs for the subject B&B, which includes the battery costs and REGD costs incurred over the planning horizon. The REGD costs are calculated by multiplying the RE generated by certain REGDs deployed at each available location l (that is, those REGDs contained in DVC_l) with their corresponding energy generation costs (that is, CST_d). Equation (2) measures the total RE generated from each location *l* during each planning period *t* ($E_{l,t}$). $E_{l,t}$ is contributed by REGDs that the model decides to deploy at l (that is, $Q_{d,l}$). Therefore, if every $Q_{d,l}$ has a zero value, $E_{l,t}$ will also be zero. Equations (3) and (4) calculate the energy saved in the batteries at the end of each planning period (I_t) , equaling the previous energy savings (I_{t-1}) increased by the RE produced in the current planning period and then decreased by the B&B's energy requirements in the same planning period. As there could be times when the total energy supply is more than the total energy consumption, W_t will capture that surplus energy when that happens. Equation (5) governs the volume of RE saved in the battery. Generally, this energy cannot exceed the battery's charge limit, which is approximately 90% of the battery's storage capacity [63]. Equation (6) governs the deployment of all REGDs suitable for each location *l*, ensuring that they do not overuse the available area at *l*. Finally, Equation (7) specifies the allowable range for each decision variable.

4. Methodology

The methodology includes analytical steps applied in a case study to enable B&Bs to make optimal decisions for their business facility, as presented in Figure 1.

As shown in Figure 1, input parameters are obtained from various sources to determine solar and wind energy data and B&B data. Then, the model is applied in two scenarios, the base case (Section 4.1) and the unconventional technology case (Section 4.2). For all scenarios, the optimization models are solved with IBM ILOG CPLEX Optimization Studio 20.10, using a personal computer powered by Intel Core i5-1035G1 processor (running at 2.4 GHz) and 16 GB RAM. Finally, an in-depth analysis of the required costs, energy flow, and emission savings is performed to analyze how the proposed system operates in both scenarios. Since many B&Bs still purchase electricity from the power grid to meet their energy demands, the base case analysis also calculates the required electricity costs from the power grid for comparison. On the other hand, some B&Bs need to use every possible location to generate sufficient RE due to limited available land. Therefore, land requirements are considered in the unconventional technology case analysis and compared to the base case.



Figure 1. Flowchart of the methodology in a case study.

4.1. The Base Case

The first scenario, the base case, aims to provide B&B owners with a good understanding of how to green their business with conventional REGDs, including solar panels and small wind turbines. Two configurations are tested, a solar-only RE and a solar-wind hybrid RE system. To gain full insight into the tradeoff between the required costs and the resulting sustainability performance, the MIP model for a solar-wind hybrid RE is performed with these different ratios: 50% RE, 60% RE, ..., and finally, 100% RE. For example, at 90% RE or 90% self-sufficiency, renewables supply 90% of the energy demand, while the rest comes from the power grid.

4.2. The Unconventional Technology Case

The second scenario, the unconventional technology case, aims to assess the possibility of deploying conventional REGDs and/or unconventional REGDs. The unconventional REGDs include the Tesla Solar Roof, the Ciel & Terre Floating Solar, and the Wind Tree. The description of each device used in this study is as follows.

The Tesla Solar Roof [62] has an aesthetic design that resembles a standard roof tile for solar energy generation. Therefore, households can use this Tesla product as a regular tile on their rooftop while also receiving clean solar energy to power their house. Another example is the Ciel & Terre Floating Solar [64], a modular floating device designed to support a single solar panel with a suitable tilted angle on the water surface. For wind energy, the Wind Tree is [65] made of dozens of small vertical-axis wind turbines mounted on a metal frame; from a distance, it looks just like a regular tree in nature. Tables A1 and A2 (in Appendix A) present the specifications of these REGDs.

4.3. Case Study of the B&B

The case study of the B&B uses a subject B&B located in Pingtung, a southern county in Taiwan. As B&Bs are a vital business for transforming rural areas into leisure businesses in Taiwan, their number increased from 40 to 7042 between 2012 and 2016 [66]. Solar and wind energy also will be the primary source of RE in Taiwan in the coming future [67].

Interestingly, Pingtung is a suitable location to produce both. Since southern Taiwan has a greater potential for solar energy than the rest of the country, Pingtung can produce approximately 263 GWh of solar energy per year [68]. Pingtung is also rich in wind energy, as 66% of the annual hourly wind speed equals and is above 3.0 m/s (the cut-in speed of a wind turbine) [69]. Moreover, Pingtung has large areas of cropland and fallow land [68] that would be ideal locations for solar/wind energy production.

4.3.1. Solar and Wind Energy Data

This study uses a meteorological report of Pingtung station from the Taiwan Central Weather Bureau, ranging from 2011 to 2020 [69]. The months of April, July, October, and January are selected to cover different seasons in Taiwan, namely spring, summer, autumn, and winter. Figure A1 (in Appendix A) summarizes the 24-h solar radiance and wind speed data in Pingtung using 10-year average data for each season-representing month. The figures also present that these hourly data can be summarized in solar/wind supply for a 6-h time period: 00:00–06:00, 06:00–12:00, 12:00–18:00, and 18:00–24:00 for the first, second, third, and fourth time periods, respectively. In order to conduct more realistic simulations of energy supply, this study considers formulas from published research [70,71] and technical specs of REGDs (Tables A1 and A2 in Appendix A).

In financial terms, RE technologies available in the current market may vary in maturity. Therefore, different approaches should be considered to provide an accurate estimate of the costs of producing and saving energy. For conventional REGDs, this study adopts the work of Ouyang and Lin [72], which calculates the LCOE based on the existing feed-in tariff. In this study, the LCOE is assumed to be 40% lower than the Taiwanese feed-in tariff [73], that is, USD 0.120/kWh for conventional solar panels and USD 0.200/kWh for wind turbines. For unconventional REGDs, this study adopts an established formula [74] that calculates LCOE based on costs, performance, and reliability. The capital costs are taken from [62,75,76], the operation and maintenance costs are assumed to be 1.5% of the capital costs, and the project's economic life is assumed to be 25 years. An interest rate of 7.0% is used since the interest rate of various RE projects ranges from 5% to 10% [77]. Accordingly, the LCOEs for unconventional REGDs are as follows: USD 0.171/kWh, USD 0.139/kWh, and USD 0.488/kWh for the Tesla Solar Roof, the Ciel & Terre Floating Solar, and the Wind Tree, respectively. For comparison, B&Bs can pay electricity from the power grid for an average of USD 0.136/kWh, depending on the timing and amount of consumption (Figure A2 in Appendix D). Following a procedure similar to unconventional REGDs and the corresponding data [63,78], the LCOE for saving energy or unit battery cost can also be calculated, that is, USD 52.72/kWh.

4.3.2. Layout at the Subject B&B

Due to difficulties in acquiring real data, this study adopts the layout of a two-story building [79] for the subject B&B in Pingtung. This B&B has a ground floor area and a total floor area of 136 and 371 m², respectively. There are a total of six rooms on the first and the second floor; four of them are guest rooms (sizes ranging from 15 to 25 m²). Outside of the main building, there is a deck, a water pond, an outdoor barbeque, a front porch, a rooftop, a balcony, and an outdoor fireplace. In addition to deploying solar panels at such conventional locations as the ground, other locations are also suitable for installation. For example, water ponds (e.g., quarry lakes, irrigation canals, and other wetlands) have been practically used to generate electricity due to the shortage of available land on the market. Figure 2 presents the graphical illustration of the subject B&B from different views, with 3D and 2D top-view. According to Figure 2b, possible locations and available areas at the subject B&B to install conventional and unconventional REGDs are listed in Table 2.



Figure 2. Graphical illustration of the subject B&B: (a) 3D; (b) 2D top-view.

Table 2. Possible locations and available areas for conventional REGDs at the subject B&B.

Locations	RFT	BLC	FRP	OBQ	DCK	OFP	WP
Area (m ²)	136	65	275	225	185	225	300
Solar panel	\checkmark						
Wind turbine	\checkmark	х	\checkmark	\checkmark	\checkmark	\checkmark	х
Tesla Solar Roof	\checkmark	х	х	х	х	х	х
Ciel & Terre Floating Solar	х	х	х	х	х	х	\checkmark
Wind Tree	х	х	\checkmark	х	х	х	x

RFT = rooftop; BLC = balcony; FRP = front porch; OBQ = outdoor barbeque; DCK = deck; OFP = outdoor fireplace; WP = water pond.

4.3.3. Demand at the Subject B&B

Due to a lack of information on actual energy profiles of buildings, especially hourly energy profiles [80], there are works of literature that use methods to derive approximate profiles. This study adopts the regression model presented by Wang [81] to create a realistic energy profile for the subject B&B. According to that study, several factors such as occupancy rate, floor area, room rate, and year of construction are included in estimating annual energy demands. Using the quantitative approaches proposed by Lamagna et al. [82] and incorporating historical monthly occupancy data from the Taiwan Tourism Bureau from 2013 to 2018 [15], the annual demands are then disaggregated into monthly ones. The representative study of hotel energy profiles by Chung and Park [83] is used to allocate the monthly demands into hourly energy demands. According to that literature, the hourly energy demand correlates with the hourly portion of the total monthly demand. This study also considers the literature that examines hourly guest activity in a day at B&B in Taiwan. As Chen et al. [84] noticed, B&B guests in Taiwan prefer to conduct outdoor activities, such as enjoying scenic places nearby. Generally, they would not stay in their room after checking in (at 15:00 or after). Consequently, the daily peak electricity demand in the B&B generally occurs during nighttime hours (i.e., the fourth time period, 18:00–24:00) when guests return from their trips. The third time period (12:00–18:00) is also a high peak demand as guests begin to occupy their rooms after check-in.

The allocation process also considers not only daily/weekly demand peaks but also holidays and general travel patterns during the year. In Taiwan, the most important holiday is the lunar new year, which occurs in January or February, and the school's winter vacation also takes place in January and February. The mid-autumn festival is another major holiday that occurs in September or October. Moreover, a spring break (four or five consecutive days) occurs at the beginning of April. Furthermore, the summer months are a prevalent time for travel. Consequently, the energy demands in June, July, and August are the highest

10 of 23

during the year. Table A3 (in Appendix B) summarizes the energy demand in each time period for each season-representing month at the subject B&B.

5. Results and Discussion

5.1. Analysis of the Base Case

5.1.1. 100% Self-Sufficiency in the Base Case

Table 3 summarizes the optimal solution for the B&B to run its business with 100% solar/wind power. In the solar-only RE system, the B&B would need to install a total of 1161 solar panels at various locations to generate sufficient energy, along with a 125-kWh battery to ensure continuous energy supply. Those solar panels would generate an average 154,977 kWh of electricity in one year. The MIP model measures the annual costs for such an RE system to be USD 25,187. For the solar–wind hybrid RE system, the B&B would need to install a total of 799 solar panels, 17 wind turbines, and a 126-kWh battery. This would produce an average 116,942 kWh of electricity each year at the cost of USD 21,499.

Table 3. The optimal solution for the base case: solar-only vs. solar-wind.

The Base	System Configuration (Unit)		Battery	Annual Energy Generation (kWh/Year)			Annual Costs (USD/Year)			Cost of	
Case	Solar Panel	Wind Turbine	(kWh)	Solar	Wind	Total	Solar	Wind	Battery	Total	USD/kWh)
Solar-only Solar-wind	1161 799	- 17	125 126	154,977 106,655	- 10,287	154,977 116,942	18,597 12,799	- 2057	6590 6643	25,187 21,499	0.1625 0.1838

Although both systems would allow the B&B to be totally independent of the power grid, their costs are also substantially higher. To be precise, when obtaining electricity only from the power grid, the B&B would need to pay USD 9059 annually (calculated from the data in Figure A2 in Appendix D), which is approximately 36% of the costs required to adopt the solar-only RE system and 42% of those required to adopt the hybrid solar–wind RE system. The results also present that both RE systems are much costlier due to overproduction, especially in the solar-only RE system. It generates more than three times the electricity than the B&B actually needs (50,217 kWh/year). This significant overproduction is due to the time difference in energy generation and consumption as well as the seasonal difference in producing solar energy.

A detailed analysis of energy flow in the solar-only RE system is performed to understand how the time difference leads to overproduction. The analysis uses energy flow in the week of July, which is considered a high-demand season for the B&B (Figure 3). The energy flow on Wednesday, the central day of the week, can be described as follows:



Figure 3. Energy generation, demand, and battery storage in a July week: solar-only scenario.

- In the first time period, t = 9 and REQ₉ = 23 kWh, no energy is produced during this period, as $E_9 = 0$. Therefore, the model allocates all energy saved in the battery during the previous period to satisfy demand, as $I_8 = 23$ kWh, and $I_9 = 0$ kWh.
- In the second time period, t = 10 and $\text{REQ}_{10} = 29$ kWh, the model allocates solar panels to generate much more energy than is required to satisfy all the demand in these periods and to save energy in the battery for later use, as $E_{10} = 253$ kWh and $I_{10} = 113$ kWh. However, some of the energy was unused in this time period, as $W_{10} = 111$ kWh.
- In the third time period, t = 11 and REQ₁₁ = 71 kWh, the model allocates solar panels to produce energy, as $E_{11} = 256$ kWh and the battery to save energy $I_{11} = 95$ kWh. The remaining energy would be unused, as $W_{11} = 204$ kWh.
- In the fourth time period, t = 12 and REQ₁₂ = 72 kWh, no energy was produced during this time period, as $E_{12} = 0$. The model allocates energy saved in the battery from the previous time period to satisfy demand, as $I_{11} = 95$ kWh, and $I_{12} = 23$ kWh.

In summary, solar panels would produce energy during daytime hours (i.e., the second and third time periods), but the B&B would consume the most energy during nighttime hours (the third and the fourth time periods). These time differences would allow sufficient surplus power to be saved in the battery to satisfy nighttime demands. Since solar radiance varies significantly from one season to the next, the 1161 solar panels (i.e., a fixed power generation capacity) certainly produce much unused energy at certain times, that is, surplus energy neither used to satisfy demand nor saved in the battery. Interestingly, the volume of unused energy exceeds the demand and occurs primarily during daytime hours. The average volume of unused energy is 10.3% higher on weekdays (Monday–Friday) than on weekends (Saturday–Sunday).

In the hybrid solar–wind system (hybrid RE), solar energy is only available during daytime hours, while wind energy may be available throughout the day. As the hybrid RE can take advantage of the complementarity between solar and wind energy, the annual costs of the hybrid RE are 17.2% lower than the annual costs of the solar-only system. Compared to prior studies (Table 1), the cost of the energy of the hybrid RE in this study (USD 1.838/kWh) is competitive, although it also depends on the cost setting of each technology.

According to Table 3, wind energy contributes 10,287 kWh (8.8%) to the annual RE generation. However, as shown in Table 4, the volume of solar and wind energy generation varies depending on the season, which affects their percentage share. For example, the percentage of wind energy for each season is as follows: 4.8%, 1.3%, 13.2%, and 22.1% in spring, summer, autumn, and winter, respectively. In the summer, solar contributes the most to RE generation (98.7%), as solar radiance reaches its maximum. In the spring and autumn, solar is still preferred, but not as much as in summer. Finally, when the wind speed reaches its maximum in the winter, the model prefers a more significant wind allocation than in other seasons. Further details on total energy generation in each time period for each season-representing month for hybrid RE are presented in Table A4 (in Appendix C).

Table 4. Energy generation (kWh) and percentage share of solar/wind for each season-representing month.

Energy (kWh)	April (Spring)	July (Summer)	October (Autumn)	January (Winter)	
Solar	9719 (95.2%)	10,853 (98.7%)	8771 (86.8%)	6988 (77.9%)	
Wind	490 (4.8%)	145 (1.3%)	1336 (13.2%)	1984 (22.1%)	

Although solar energy completely dominates RE generation in Pingtung, the model still relies on wind energy to simultaneously supply power. In addition to seasonal variations, solar is only available during daytime hours and wind may be available throughout the day. Considering B&Bs receive energy supply from RE that depends on local weather and B&Bs also have to cope with the fluctuations in energy demand due to peak periods/seasons, determining the specific size of the batteries becomes a critical decision. However, there are significant differences between the LCOEs for generating and saving energy (Section 4.3.1). This condition leads to different decisions regarding surplus energy. Sometimes it is more cost-effective to save that surplus for later use, resulting in greater battery capacity and higher annual costs. However, it is more cost-effective to let it be unused at other times.

In terms of emissions, most of Taiwan's electricity generation is provided by Taipower, a state-owned power company that uses fossil fuels to generate electricity. However, generating one kWh of electricity from the power grid is also responsible for 0.532 kgCO₂e (kilograms of carbon dioxide equivalent) [85]. If the B&B is able to run its business with 100% RE, this would result in the avoidance of CO₂ emissions of 26,715 kgCO₂e or 26.7 tCO₂e (tons of carbon dioxide equivalent) per year. Compared to per capita CO₂ emissions in Taiwan (11.02 tCO₂e/person) [86], the reduction may appear insignificant. However, if half of Taiwanese B&Bs switch their primary energy sources to RE, they could potentially reduce emissions by 94,010 tCO₂e per year. It is worth mentioning that there are approximately 7042 B&Bs in Taiwan [66].

Taiwan is still in the process of formulating appropriate policies to govern CO₂ emissions. Therefore, the following analysis is based on the average carbon tax rate in Europe, that is, USD 42.29/tCO₂e [87]. Consequently, if the B&B purchases electricity from the power grid, the actual annual electricity bill, including emission costs, would be USD 10,188. This is about half of the costs of hybrid RE (USD 21,499). However, electricity in Taiwan has been unreasonably cheap for many years because the prices do not reflect the actual costs. For example, the average electricity price in Taiwan is USD 0.136/kWh (Figure A2 in Appendix D) compared to some developed countries, such as Germany (USD 0.323/kWh) and the UK (USD 0.251/kWh) [88]. The Taiwanese government has intentionally controlled electricity through highly subsidized prices. The burden is shouldered by Taipower, whose electricity business has suffered significant losses [89]. If further rational pricing of electricity is considered, the cost difference between the hybrid RE and the power grid would not be significantly different.

5.1.2. Different Self-Sufficiency Ratios in the Base Case

To ensure the model provides an accurate solution to achieve a specific ratio of selfsufficiency, an equation about demand would be added as in (8).

$$REQN_t \le REQ_t \quad \forall t \in T$$
 (8)

where $REQN_t$ is a decision variable representing the actual energy demand, which needs to be satisfied with RE during the time period *t*. Each $REQN_t$ must be less than or equal to REQ_t which is the (real) energy demand at the subject B&B during planning period *t*.

The analysis of the energy flow at different self-sufficiency ratios is performed to understand how RE generation and storage can satisfy demand in the week of July (Figure 4). The energy flow on Wednesday as the central day of the week at self-sufficiency 90% can be described as follows:

- In the first time period, t = 9 and REQ₉ = 23 kWh, the model allocates the power grid to satisfy the demand as $REQN_9 = 0$ and there is no saving energy in the battery from the previous time period, $I_8 = 0$.
- In the second and third time period, t = 10 and t = 11, the model allocates solar and wind energy to satisfy all demand in these periods, as $REQN_{10} = 29$ kWh and $REQN_{11} = 71$ kWh. However, there is some surplus energy that would be saved in the battery, $I_{10} = 71$ kWh and $I_{11} = 46$ kWh.
- In the fourth time period, t = 12, and $\text{REQ}_{12} = 46$ kWh, the model allocates energy saved in the battery from the previous time period to satisfy demand as $REQN_{12} = 0$, $I_{11} = 46$ kWh, and $I_{12} = 0$.



Figure 4. Energy generation and storage in a July week: (a) 90% self-sufficiency; (b) 100% self-sufficiency.

On the other hand, at 100% self-sufficiency, the model would allocate solar and wind in daytime hours to generate much more energy than required. The energy generation is not only used to satisfy demand at that period but also to save for the next period. For example, in the third time period, t = 11 and $\text{REQ}_{t=11} = 71$ kWh, the model would allocate $E_{t=11} = 179$ kWh and $I_{t=11} = 95$ kWh. As a result, the volume of energy stored in the battery increases significantly since the model decides to generate more than twice as much energy at 90% self-sufficiency (Figure 4b).

Considering different ratios of self-sufficiency, Figure 5 presents the annual costs from 50% to 100% sufficiency with a difference of 10%. As a variety of self-sufficiency is evaluated, the annual costs also consider the costs of the power grid and emissions. The annual costs increase depending on the ratio of self-sufficiency as follows: USD 11,044, USD 12,360, USD 15,019, USD 15,084, USD 15,797, and USD 21,499 for self-sufficiency ratios of 50%, 60%, 70%, 80%, 90%, and 100%, respectively (Figure 5a). There is a tradeoff between the annual costs and the sustainability performance that needs to be analyzed. Since increasing ratios affect costs differently, 50% self-sufficiency can be used as a baseline. For example, if the B&B chooses to be 90% self-sufficient instead of 50%, the annual costs would increase by 42.6%. Emission costs, on the other hand, decrease by 80.0% as the power grid supplies less electricity, from 25,108 kWh to 5022 kWh per year.





Figure 5b summarizes the tradeoff between the annual and emission costs when different ratios are considered. The annual costs would increase by 11.8%, 35.7%, 36.2%, 42.6%, and 94.0% (black bars) if the ratio of self-sufficiency increases to 60%, 70%, 80%, 90%, and 100%, respectively. However, the B&B achieves a 20% to 100% reduction in emission costs (green bars) when increasing the ratio. Therefore, B&B owners can choose the most suitable energy transition tradeoff based on the self-sufficiency ratio as the target to be achieved.

Figure 5 also presents that a slight increase from 90% to 100% self-sufficiency would lead to a huge jump in the annual costs, from USD 15,797 to USD 21,499 (more than 35%). In fact, the MIP model decides on the fixed power generation capacity that ensures an uninterrupted supply for a whole year. However, as described in Figure 4, the model would behave completely differently depending on the self-sufficiency to cope with time differences between energy generation and consumption. These considerations lead to significant differences in battery capacity, energy generation, and costs when comparing 90% to 100% self-sufficiency, as summarized in Table 5. As solar energy increases in terms of the annual volume and percentage of energy sources, more capacity is required to store energy. At 100% self-sufficiency, the storage capacity reaches 126 kWh, more than double the capacity at 90% self-sufficiency (51 kWh). It is obvious that the battery costs also increase significantly, from USD 2689 to USD 6643.

Table 5. Comparison of self-sufficiency 90% vs. 100%: battery capacity, energy generation, costs.

Self-Sufficiency	Battery Capacity	Annual Ener	gy Generation	Annual Costs (USD /Year)				
Ratio	(kWh)	Solar	Wind	Total	Solar	Wind	Battery	Total
90%	51	46,854	33,655	80,509	5622	6731	2689	15,042
100%	126	106,655	10,287	116,942	12,799	2057	6643	21,499

According to Table 5, the MIP model decides to generate more energy (50,217 kWh) than is required in either ratio. However, annual energy generation is less at 90% than 100% self-sufficiency (80,509 kWh vs. 116,942 kWh). As a result, the volume of the unused energy at 90% self-sufficiency is approximately 45% of the unused energy at 100% self-sufficiency. All these findings further reveal how a slight increase in self-sufficiency from 90% to 100% leads to significant changes in the annual costs (excluding the power grid costs and emission costs). Due to more affordable investments, incorporating the power grid into the RE system through a certain ratio of self-sufficiency would allow B&Bs to have better access to the energy transition. However, they still need to consider the emissions tradeoff that results from this integration.

To highlight the role of the power grid, Figure 6 displays the satisfaction of energy demand by incorporating the power grid into the RE system at a self-sufficiency of 70% and 90%. As seen in both figures, the energy demands are significantly higher in the summer months than winter months. However, given a specific power generation capacity to produce RE, there will certainly be times in which that capacity is fully utilized, and there will also be times in which some capacity is not used. Consequently, it is important to minimize this unused capacity as appropriately as possible. When there is an option to acquire electricity from the power grid, this electricity would have the effect of smothering the remaining power demand to be satisfied by RE. The green bars in Figure 6a,b demonstrate such a smothering effect for the remaining power demand satisfied by RE, and that it is much more smooth at a self-sufficiency of 70% (Figure 6a) than 90% (Figure 6b).

In summary, allowing a certain demand to be satisfied by the power grid would utilize RE capacity more evenly and reduce unused capacity, which in turn improves investment utilization.



Figure 6. Satisfaction of energy demand by the power grid and RE: (**a**) 70% self-sufficiency; (**b**) 90% self-sufficiency.

Figure 7 summarizes all the details of the battery installation in the RE system. It is noticeable that the battery is required when the self-sufficiency is between 80% and 100%, and no battery when the self-sufficiency is between 50% and 70% (Figure 7a). Taking July as an example, solar and wind energy is only available during daytime hours, as wind speed is low at night until early morning. However, the B&B has a peak load during nighttime hours. Therefore, the model can utilize both with and without batteries in the system to balance the fluctuations. Figure 7b illustrates the highly dynamic demand/supply in the B&B as a ratio between the total volume of RE generation and energy demand for an entire year. For comparison, the analysis is performed either with batteries (at 100% self-sufficiency) or without batteries (at 50% self-sufficiency).



Figure 7. Battery installation: (**a**) battery capacity for different self-sufficiency ratios; (**b**) comparison with or without batteries in the system.

At 100% self-sufficiency, Figure 7b demonstrates huge fluctuations throughout the year. The integration of solar and wind is not sufficient to cope with these fluctuations. Therefore, the battery must ensure an adequate balance between demand and supply in the RE system. On the other hand, it is also possible that the MIP model does not include the battery in the system. At 50% self-sufficiency, the power grid would play a role in achieving an adequate balance between energy demand and supply. Therefore, the remaining fluctuations become more smooth, as presented by the dashed line in Figure 6b. The integration of solar and wind is then able to balance these fluctuations without the battery's support. Overall, the annual costs would decrease significantly if the system does not require batteries. The saving without the battery is USD 10,455 per year, nearly half the cost of an RE system with

the battery (USD 21,499). Due to significant cost savings, the battery has an important role in achieving 100% RE IN B&Bs.

5.2. Analysis of the Unconventional Technology Case

Due to the problem with the cost minimization objective and the higher cost of unconventional REGDs, the MIP model would naturally prefer conventional solar panels and wind turbines. Therefore, to ensure that the MIP model adopts some unconventional REGDs, $d \in DVC_l$ in constraint (6) (Section 3) is adjusted so that DVC_l only contains specific REGDs for certain locations. For example, if *l* is the rooftop, DVC_l becomes {Tesla Solar Roof} (in the base case it was {solar panel, wind turbine}). As another example, DVC_l is the Wind Tree when *l* is the front porch; DVC_l is the Ciel & Terre Floating Solar when *l* is the water pond. On the other hand, conventional solar panels and wind turbines are still available for other locations. Table 6 lists all possible locations and areas where conventional and unconventional REGDs can be deployed at the subject B&B.

Table 6. Possible locations and available areas for various REGDs at the subject	B&B.
--	------

Locations	RFT	BLC	FRP	OBQ	DCK	OFP	WP
Area (m ²)	136	65	275	225	185	225	300
Solar panel	х	\checkmark	х	\checkmark	\checkmark	\checkmark	х
Wind turbine	х	х	х	\checkmark	\checkmark	\checkmark	х
Tesla Solar Roof	\checkmark	х	х	х	х	х	х
Ciel & Terre Floating Solar	х	х	х	х	х	х	\checkmark
Wind Tree	x	х	\checkmark	х	х	х	x

RFT = rooftop; BLC = balcony; FRP = front porch; OBQ = outdoor barbeque; DCK = deck; OFP = outdoor fireplace; WP = water pond.

Using these data adjustments as described above, Table 7 contains the optimal solution for the unconventional technology case. The B&B would need to install 328 solar panels, 7 wind turbines, 81 units of the Tesla Solar Roof, 250 units of the Ciel Terre Floating Solar, 1 unit of the Wind Tree, and a 123-kWh battery. This would produce an average 114,474 kWh of electricity per year at the cost of USD 24,670.

System Configuration B (Unit) Ca			Battery Capacity	An Genera	Annual Energy Generation (kWh/Year)			Annual Costs (USD/Year)				
SP	WT	TSL	C&T	WTR	(kWh)	Solar	Wind	Total	Solar	Wind	Battery	Total
328	7	81	250	1	123	103,738	10,736	114,474	14,158	4028	6484	24,670

Table 7. The optimal solution for the unconventional technology case.

SP = solar panel; WT = wind turbine; TSL = Tesla Solar Roof; C&T = Ciel & Terre Floating Solar; WTR = Wind Tree.

5.3. Comparison of the Base Case and the Unconventional Technology Case

From the data in Tables 3 and 7, which present the optimal solution for the base and the unconventional technology case, respectively, the following conclusions can be drawn:

- Both scenarios generate more energy than the B&B requires (50,217 kWh/year). Since the Wind Tree has higher efficiency than a wind turbine, wind energy can be produced even though the wind speed is below 3.0 m/s at certain time periods. This capability increases the annual volume and share of wind energy in the RE system, from 10,287 kWh/year (8.8%) in the base case to 10,736 kWh/year (9.4%) in the unconventional technology case.
- As the share of wind energy increases in the unconventional technology case, the complementarity between solar and wind energy in the RE system also increases, resulting in a decrease in the volume of unused energy in the system. The volume of unused energy in the unconventional technology case is 3.7% less than in the base case.

- The increasing complementarity also affects the battery capacity in the system, from 126 kWh to 123 kWh in the base case and the unconventional technology case, respectively.
- The annual costs of the unconventional technology case are 14.7% higher than in the base case. However, fewer solar REGDs are required in the unconventional technology case than in the base case (799 units vs. 659 units), affecting the area required to install solar REGDs (927 m² vs. 796 m²).
- The required area to install REGDs in the unconventional technology case is 873 m², which is 14.2% less than in the base case.

Overall, the comparison between the base and the unconventional technology case is listed in Table 8

Table 8. Comparison between the base case and unconventional technology case.

Comparison	Base Case	Unconventional Technology Case
Energy generation (kWh/year)	116,942	114,474
Unused energy (kWh/year)	66,725	64,257
Annual costs (USD/year)	21,499	24,670
Area required (m^2)	1017	873

6. Conclusions

This study aims to provide a practical approach for all B&B businesses to make optimal location-sizing decisions on deploying hybrid RE technologies. An MIP model is developed and applied in a case study with realistic B&B settings and energy data. The case study contains two scenarios, the base case and the unconventional technology case. The MIP model successfully solves both scenarios to optimality, identifying an optimal RE system that can generate an annual 116,942 kWh of RE for the base case (at the cost of USD 21,499) and an annual 114,474 kWh of RE for the unconventional technology case (at the cost of USD 24,670), respectively. This would result in an emissions saving of $26,715 \text{ kgCO}_2\text{e}/\text{year}$ (compared to purchasing the required electricity from the power grid). Although the annual cost of hybrid RE is 14.6% less than solar-only, it is still significantly higher than the annual cost of the power grid. In addition to the cost difference between RE technologies and the power grid, overproduction due to time differences in energy generation and energy consumption also contributes to the high cost of hybrid RE. Therefore, simulating the energy transition by sharing the energy demand between RE and the power grid (based on a specific share) would allow B&Bs to initiate the energy transition with affordable costs and acceptable emissions. With 50% self-sufficiency as the baseline, annual costs would increase from 11.8% to 94.0% if self-sufficiency is increased from 60% to 100%. The annual costs in the unconventional technology case are 14.7% higher than in the base case. However, the area needed to install REGDs in the unconventional technology case is 14.2% less than the base case.

As B&Bs can make smart decisions in the RE mix, the performance reliability of REGDs becomes critical as they are exposed to outdoor conditions. Therefore, considering age-related performance degradation on RE technologies would enhance the value of future research. Future studies may also investigate emission reduction targets as, in reality, the maximum emissions allowed by policymakers become a critical challenge in energy planning.

Author Contributions: Conceptualization, C.-T.W. and Y.-J.H.; methodology, C.-T.W. and Y.-J.H.; software, L.T.; validation, C.-T.W. and L.T.; formal analysis, C.-T.W. and L.T.; investigation, L.T.; resources, C.-T.W.; data curation, C.-T.W.; writing—original draft preparation, C.-T.W. and L.T.; writing—review and editing, C.-T.W. and L.T.; visualization, L.T.; supervision, C.-T.W.; project administration, C.-T.W.; funding acquisition, C.-T.W. All authors have read and agreed to the published version of the manuscript.

Funding: This research received financial support from two organizations at National Central University: The Office of International Affairs and the Graduate Institute of Industrial Management.

Institutional Review Board Statement: Not applicable.

Informed Consent Statement: Not applicable.

Data Availability Statement: This study only uses publicly available data, with sources all clearly indicated in the paper.

Acknowledgments: The authors wish to express a special thanks to Aiyuan Homestay, a family-run B&B situated in Fangyuan Township, Changhua County, Taiwan, for their generous sharing of practical experience in onsite renewable energy generation.

Conflicts of Interest: The authors declare no conflict of interest.

Subscripts and Sets

Т	set of all planning periods in the planning horizon of the MIP model (indexed by <i>t</i> ,
	that is, <i>t</i> represents any of the planning periods in T)
L	set of all the locations at the B&B in which certain REGDs can be deployed (indexed by <i>l</i>)
DVC _l	set of all REGDs that can be deployed in location <i>l</i> (indexed by <i>d</i>)

Parameters

ARA_l	total available area of location <i>l</i>
UNT _d	the area that each unit of <i>d</i> requires to work properly
REQ_t	the B&B's expected energy demands (kWh) in planning period t
ENR _{d,t}	the expected RE (kWh) that each unit of <i>d</i> can produce during planning period <i>t</i>
BCL	energy storage capacity that each battery can provide
CST_d	LCOE, that is, the levelized cost of energy, for using <i>d</i> to produce RE (USD/kWh)
CBT	unit battery costs (USD/kWh)

Decision Variables

 $Q_{d,l}$ total units of *d* that the B&B will deploy at location *l*.

Appendix A. Preparing Solar and Wind Energy Data

Table A1 provides specifications for solar REGDs, such as a conventional solar panel [90], the Tesla Solar Roof [91], and the Ciel & Terre Floating Solar [75,92]. Table A2 contains specifications for wind REGDs, such as a wind turbine [93] and the Wind Tree [94], which consider spacing between turbines based on rotor diameter [95].

Table A1. Technical specifications for solar REGDs.

Solar REGD	Specific Type	Max Power (W)	Efficiency (%)	Area (m ² /Panel)
Solar panel	Sanyo HIT-200BA19	200	17.2	1.16
Tesla Solar Roof	Tesla SC-330	330	19.7	1.67
Ciel & Terre Floating Solar	Hydrelio Air 1440 12°	295	18.1	1.12

Table A2. Technical specifications for wind REGDs.

Wind REGD	Specific Type	Max Power (W)	Cut-in/Rated/Cut-Out (m/s)	Diameter (m)	Area (m ² /Unit)	
Wind turbine	Atlantis WB 20	300	3/10/none	1.5	5.30	
Wind Tree	New Wind Tree 36A	5868	2/12/18	7.9	40.50	

Figure A1 summarizes the hourly solar radiance and wind speed in Pingtung using 10-year average data from 2011 to 2020 [69] for each season-representing month. It is noticeable that wind speed must be 3.0 m/s (cut-in speed) or higher for wind turbines to generate energy; otherwise, wind energy is unavailable.



Figure A1. The 24-h data in Pingtung (a) solar radiance; (b) wind speed.

Appendix B. Preparing Energy Demand at the Subject B&B

Table A3 summarizes the average energy demand (kWh) in each time period for each season-representing month. This table presents a side-by-side comparison of the demand for one day on a weekday (Monday–Friday) with the demand for one day on a weekend (Saturday–Sunday).

Table A3. Average energy demand (kWh) in each time period for each season-representing month at the subject B&B.

Daily Time Periods	April (Spring)		July (Summer)		October (Autumn)		January (Winter)	
	Weekday	Weekend	Weekday	Weekend	Weekday	Weekend	Weekday	Weekend
00:00-06:00	15.6	18.7	23.3	27.9	14.8	17.8	14.0	16.8
06:00-12:00	19.5	23.4	29.1	35.0	18.6	22.3	17.5	21.0
12:00-18:00	47.5	57.0	71.0	85.2	45.3	54.4	42.7	51.2
18:00-24:00	48.3	58.0	72.2	86.6	46.0	55.3	43.4	52.1

Appendix C. Determining Optimal Solution at the Subject B&B

Table A4 summarizes the optimal solution of total solar/wind energy generation (kWh) for the base case using conventional REGDs (conventional solar panels and wind turbines). This table supports the details of Table 4 as it is presented in each time period for each season-representing month.

Table A4. Total solar/wind energy generation (kWh) for hybrid RE (base case) in each time period for each season-representing month at the subject B&B.

Daily Time Periods	April (Spring)		July (Summer)		October (Autumn)		January (Winter)	
	Solar	Wind	Solar	Wind	Solar	Wind	Solar	Wind
00:00-06:00	0	96	0	0	0	305	0	460
06:00-12:00	5061	139	5400	70	4884	363	3494	537
12:00-18:00	4658	147	5453	75	3887	338	3494	505
18:00-24:00	0	108	0	0	0	330	0	482

Appendix D. Preparing the Costs of Purchasing Electricity from the Power Grid

Taipower, the primary authority for electricity generation and transmission in Taiwan, uses progressively increasing rates based on monthly power consumption for commercial enterprises. Figure A2 presents this electricity tariff, which provides different rates for summer and non-summer [96]. According to this figure, prices are higher in the summer months (June to September) than in the other months (October to May). The purpose of



this policy is to save more energy, especially in summer, as people tend to consume more electricity in this season.

Figure A2. Monthly electricity tariff from the power grid in Taiwan.

References

- 1. World Meteorological Organization. State of Global Climate 2020; WMO: Geneva, Switzerland, 2020.
- 2. International Energy Agency (IEA). CO₂ Emissions from Fuel Combustion; IEA: Paris, France, 2019.
- 3. UNWTO. International Tourism Highlights; UNWTO: Madrid, Spain, 2020.
- 4. Lenzen, M.; Sun, Y.Y.; Faturay, F.; Ting, Y.P.; Geschke, A.; Malik, A. The carbon footprint of global tourism. *Nat. Clim. Chang.* 2018, *8*, 522–528. [CrossRef]
- 5. Scott, D. Sustainable Tourism and the Grand Challenge of Climate Change. Sustainability 2021, 13, 1966. [CrossRef]
- Roxas, F.M.Y.; Rivera, J.P.R.; Gutierrez, E.L.M. Mapping stakeholders' roles in governing sustainable tourism destinations. *J. Hosp. Tour. Manag.* 2020, 45, 387–398. [CrossRef]
- United Nations Environment Programme & World Tourism Organization. Making Tourism More Sustainable: A Guide for Policy Makers; United Nations Environment Programme & World Tourism Organization: Madrid, Spain, 2005.
- 8. Scott, D.; Gössling, S. *Tourism and Climate Change Mitigation Embracing the Paris Agreement*; European Travel Commission, Ed.; European Travel Commission: Brussel, Belgium, 2018.
- 9. Sustainable Development. Available online: https://www.unwto.org/sustainable-development (accessed on 5 October 2021).
- 10. Tsai, K.; Lin, T.; Hwang, R.; Huang, Y. Carbon dioxide emissions generated by energy consumption of hotels and homestay facilities in Taiwan. *Tour. Manag.* **2014**, *42*, 13–21. [CrossRef]
- 11. Hu, A.H.; Huang, C.Y.; Chen, C.F.; Kuo, C.H.; Hsu, C.W. Assessing carbon footprint in the life cycle of accommodation: The case of an international tourist hotel. *Int. J. Sustain. Dev. World Ecol.* **2015**, *22*, 313–323. [CrossRef]
- 12. UNWTO & UNEP. Climate Change and Tourism: Responding to Global Challenges; UNWTO & UNEP: Madrid, Spain, 2008.
- Warren, C.; Becken, S.; Nguyen, K.; Stewart, R. Transitioning to smart sustainable tourist accommodation: Service innovation results. J. Clean. Prod. 2018, 201, 599–608. [CrossRef]
- 14. Kuo, C.M.; Tseng, C.Y.; Chen, L.C. Choosing between exiting or innovative solutions for B&Bs. *Int. J. Hosp. Manag.* **2018**, *73*, 12–19.
- 15. Monthly Report of Home Stay Facilities Operations in Taiwan. Available online: https://admin.taiwan.net.tw/ (accessed on 24 June 2021).
- 16. Northern Ireland Guesthouse, Bed & Breakfast, and Guest Accommodation Occupancy Rates. Available online: https://www.nisra.gov.uk/statistics/tourism/hotel-occupancy-surveys (accessed on 4 April 2022).
- 17. World Bank Group. Tourism and the Sharing Economy: Policy & Potential of Sustainable Peer-to-Peer Accommodations; World Bank Group: Washington, DC, USA, 2018.
- Long, F.; Liu, J.; Zhang, S.; Yu, H.; Jiang, H. Development characteristics and evolution mechanism of homestay agglomeration in Mogan Mountain, China. *Sustainability* 2018, 10, 2964. [CrossRef]
- 19. Pereira, V.; Silva, G.M. Sustainability Practices in Hospitality: Case study of a luxury hotel in Arrabida Natural Park. *Sustainability* **2021**, *13*, 3164. [CrossRef]
- Abdou, A.H.; Hassan, T.H.; Dief, E.; Moustafa, M. A Description of green hotel practices and their role in achieving sustainable development. *Sustainability* 2020, 12, 9624. [CrossRef]
- 21. IRENA. A Path to Prosperity: Renewable Energy for Islands; IRENA: Abu Dhabi, United Arab Emirates, 2016.

- 22. Okkonen, L.; Lehtonen, O. Socio-economic impacts of community wind power projects in Northern Scotland. *Renew. Energy* 2016, 85, 826–833. [CrossRef]
- 23. Moise, M.S.; Gil-Saura, I.; Ruiz Molina, M.E. The importance of green practices for hotel guests: Does gender matter? *Econ. Res. Istraz.* **2021**, *34*, 3508–3529. [CrossRef]
- Guttentag, D.; Smith, S.; Potwarka, L.; Havitz, M. Why tourists choose Airbnb: A motivation-based segmentation study. J. Travel Res. 2018, 57, 342–359. [CrossRef]
- 25. So, K.K.F.; Kim, H.; Oh, H. What makes Airbnb experiences enjoyable? the effects of environmental stimuli on perceived enjoyment and repurchase intention. *J. Travel Res.* **2021**, *60*, 1018–1038. [CrossRef]
- Zhu, Y.; Cheng, M.; Wang, J.; Ma, L.; Jiang, R. The construction of home feeling by Airbnb guests in the sharing economy: A semantics perspective. *Ann. Tour. Res.* 2019, 75, 308–321. [CrossRef]
- 27. Tang, T. Competing through customer social capital: The proactive personality of bed and breakfast operators. *Asia Pac. J. Tour. Res.* **2015**, *20*, 133–151. [CrossRef]
- 28. International Energy Agency. Global EV Outlook 2020; International Energy Agency: Paris, France, 2020.
- 29. Marriot International Electric Vehicle Charging Stations. Available online: https://www.marriott.com/corporate-social-responsibility/electric-vehicle-hotels.mi (accessed on 5 October 2021).
- 30. Hotels Race to Fill Growing Demand for Electric Car Chargers. Available online: https://www.hotelmanagement.net/tech/ hotels-drive-revenue-by-charging-electric-cars (accessed on 5 October 2021).
- Zhao, J.; Carney, M.; Zhang, S.; Zhu, L. How does an intra-family succession effect strategic change and performance in China's family firms? *Asia Pac. J. Manag.* 2020, 37, 363–389. [CrossRef]
- 32. SolarPower Europe. Global Market Outlook for Solar Power 2018–2022; SolarPower Europe: Brussels, Belgium, 2018.
- Greenview. Green Lodging Trends Report 2018; Greenview: Singapore, 2019. Available online: https://greenview.sg/services/ green-lodging-trends-report/ (accessed on 27 September 2021).
- Sardianou, E.; Kostakis, I. Perceived barriers to invest in renewable energy sources in the Cretan hotel industry. *Int. J. Sustain.* Energy 2020, 39, 240–249. [CrossRef]
- 35. Ye, S.; Xiao, H.; Zhou, L. Small accommodation business growth in rural areas: Effects on guest experience and financial performance. *Int. J. Hosp. Manag.* **2019**, *76*, 29–38. [CrossRef]
- Casteleiro-Roca, J.L.; Gómez-González, J.F.; Calvo-Rolle, J.L.; Jove, E.; Quintián, H.; Diaz, B.G.; Perez, J.A.M. Short-term energy demand forecast in hotels using hybrid intelligent modeling. *Sensors* 2019, 19, 2485. [CrossRef]
- 37. Anoune, K.; Bouya, M.; Astito, A.; Abdellah, A. Ben. Sizing methods and optimization techniques for pv-wind based hybrid renewable energy system: A review. *Renew. Sustain. Energy Rev.* **2018**, *93*, 652–673. [CrossRef]
- Gross, S. Renewables, Land Use, and Local Opposition in the United States Executive Summary. Available online: https: //www.brookings.edu/wpcontent/uploads/2020/01/FP_20200113_renewables_land_use_local_opposition_gross.pdf (accessed on 26 October 2021).
- Sadollah, A.; Nasir, M.; Geem, Z.W. Sustainability and optimization: From conceptual fundamentals to applications. *Sustainability* 2020, 12, 2027. [CrossRef]
- 40. Warren, C.; Becken, S. Saving energy and water in tourist accommodation: A systematic literature review (1987–2015). *Int. J. Tour. Res.* **2017**, *19*, 289–303. [CrossRef]
- 41. Ammari, C.; Belatrache, D.; Touhami, B.; Makhloufi, S. Sizing, Optimization, Control and Energy Management of Hybrid Renewable Energy System—A Review. *Energy Built Environ.* **2021**, *in press.* [CrossRef]
- Ganguly, P.; Kalam, A.; Zayegh, A. Solar-wind hybrid renewable energy system: Current status of research on configurations, control, and sizing methodologies. In *Hybrid-Renewable Energy Systems in Microgrids: Integration, Developments and Control*; Fathima, A.H., Prabaharan, N., Palanisamy, K., Kalam, A., Mekhilef, S., Justo, J.J., Eds.; Woodhead Publishing Series in Energy: Cambridge, UK, 2018; Volume 12, pp. 219–248.
- 43. HOMER Pro. Available online: https://www.homerenergy.com/products/pro/features.html (accessed on 2 February 2022).
- 44. Uwineza, L.; Kim, H.G.; Kim, C.K. Feasibilty study of integrating the renewable energy system in Popova island using the Monte Carlo model and HOMER. *Energy Strateg. Rev.* **2021**, *33*, 100607. [CrossRef]
- Shin, Y.; Yong, W.; Hyung, T.; Jung, S.; Kim, H. Capacity design and operation planning of a hybrid PV-wind-battery-diesel power generation system in the case of Deokjeok Island. *Appl. Therm. Eng.* 2015, *89*, 514–525. [CrossRef]
- 46. Lauinger, D.; Caliandro, P.; Van Herle, J.; Kuhn, D. A linear programming approach to the optimization of residential energy systems. *J. Energy Storage* **2016**, *7*, 24–37. [CrossRef]
- Dalton, G.J.; Lockington, D.A.; Baldock, T.E. Feasibility analysis of stand-alone renewable energy supply options for a large hotel. *Renew. Energy* 2008, 33, 1475–1490. [CrossRef]
- Dalton, G.J.; Lockington, D.A.; Baldock, T.E. Case Study feasibility analysis of renewable energy supply options for small to medium-sized tourist accommodations. *Renew. Energy* 2009, 34, 1134–1144. [CrossRef]
- Güler, Ö.; Akda, S.A.; Dinc, M.E. Feasibility analysis of medium-sized hotel's electrical energy consumption with hybrid systems. Sustain. Cities Soc. 2013, 9, 15–22. [CrossRef]
- 50. Fazelpour, F.; Soltani, N.; Rosen, M.A. Feasibility of satisfying electrical energy needs with hybrid systems for a medium-size hotel on Kish island, Iran. *Energy* 2014, *73*, 856–865. [CrossRef]

- 51. Shezan, S.K.A.; Saidur, R.; Ullah, K.R.; Hossain, A.; Chong, W.T.; Julai, S. Feasibility analysis of a hybrid off-grid wind–DG-battery energy system for the eco-tourism remote areas. *Clean Technol. Environ. Policy* **2015**, *7*, 2417–2430. [CrossRef]
- Khan, M.R.; Jidin, R.; Pasupuleti, J.; Shaaya, S. Optimal combination of solar, wind, micro-hydro and diesel systems based on actual seasonal load profiles for a resort island in the South China Sea. *Energy* 2015, 82, 80–97. [CrossRef]
- 53. Hossain, M.; Mekhilef, S.; Olatomiwa, L. Performance evaluation of a stand-alone PV-wind-diesel-battery hybrid system feasible for a large resort center in South China Sea, Malaysia. *Sustain. Cities Soc.* **2017**, *28*, 358–366. [CrossRef]
- 54. Borowski, M.; Mazur, P.; Kleszcz, S.; Zwolińska, K. Energy monitoring in a heating and cooling system in a building based on the example of the Turówka hotel. *Energies* **2020**, *13*, 1968. [CrossRef]
- 55. TRNSYS 18. Available online: https://sel.me.wisc.edu/trnsys/ (accessed on 26 October 2021).
- 56. Al Ka'bi, A.H. Comparison of energy simulation applications used in green building. *Ann. Telecommun.* **2020**, *75*, 271–290. [CrossRef]
- 57. Shrivastava, R.L.; Kumar, V.; Untawale, S.P. Modeling and simulation of solar water heater: A TRNSYS perspective. *Renew. Sustain. Energy Rev.* **2017**, *67*, 126–143. [CrossRef]
- 58. Buonomano, A.; Calise, F.; Palombo, A.; Vicidomini, M. Energy and economic analysis of geothermal–solar trigeneration systems: A case study for a hotel building in Ischia. *Appl. Energy* **2015**, *138*, 224–241. [CrossRef]
- 59. Beccali, M.; Finocchiaro, P.; Ippolito, M.G.; Leone, G.; Panno, D.; Zizzo, G. Analysis of some renewable energy uses and demand side measures for hotels on small Mediterranean Islands: A case study. *Energy* **2018**, *157*, 106–114. [CrossRef]
- 60. Soheyli, S.; Shafiei Mayam, M.H.; Mehrjoo, M. Modeling a Novel CCHP system including solar and wind renewable energy resources and sizing by a CC-MOPSO algorithm. *Appl. Energy* **2016**, *184*, 375–395. [CrossRef]
- 61. Meschede, H.; Dunkelberg, H.; Stöhr, F.; Peesel, R.; Hesselbach, J. Assessment of probabilistic distributed factors influencing renewable energy supply for hotels using Monte-Carlo methods. *Energy* **2017**, *128*, 86–100. [CrossRef]
- 62. Tesla Solar Roof. Available online: https://www.tesla.com/solarroof (accessed on 27 October 2021).
- 63. Tesla Powerwall 2 AC Datasheet—United Kingdom. Available online: https://www.tesla.com/sites/default/files/pdfs/powerwall/Powerwall%202_AC_Datasheet_en_GB.pdf (accessed on 26 October 2021).
- 64. Ciel & Terre. Available online: https://www.ciel-et-terre.net/ (accessed on 27 October 2021).
- 65. New World Wind. Available online: https://newworldwind.com/en/wind-tree/ (accessed on 27 September 2021).
- Chiu, C.N. How can managerial efficiency be improved? Evidence from the bed and breakfast industry. *Tour. Manag. Perspect.* 2018, 27, 111–124. [CrossRef]
- 67. Chen, H.H.; Lee, A.H.I. Comprehensive overview of renewable energy development in Taiwan. *Renew. Sustain. Energy Rev.* 2014, 37, 215–228. [CrossRef]
- 68. Kung, C.; Mccarl, B.A. The potential role of renewable electricity generation in Taiwan. Energy Policy 2020, 138, 111227. [CrossRef]
- 69. Data Bank for Atmospheric & Hydrologic Research—Taiwan Meteorological Data. Available online: https://dbar.pccu.edu.tw/ (accessed on 24 June 2021).
- Mazidi, M.; Zakariazadeh, A.; Jadid, S.; Siano, P. Integrated scheduling of renewable generation and demand response programs in a microgrid. *Energy Convers. Manag.* 2014, 86, 1118–1127. [CrossRef]
- Mostafaeipour, A.; Sedaghat, A.; Dehghan-Niri, A.A.; Kalantar, V. Wind energy feasibility study for city of Shahrbabak in Iran. *Renew. Sustain. Energy Rev.* 2011, 15, 2545–2556. [CrossRef]
- Ouyang, X.; Lin, B. Levelized cost of electricity (LCOE) of renewable energies and required subsidies in China. *Energy Policy* 2014, 70, 64–73. [CrossRef]
- 73. Feed-In Tariffs of Renewable Energy in Taiwan 2020—Bureau of Energy. Available online: https://www.moeaboe.gov.tw (accessed on 26 February 2021).
- Simple Levelized Cost of Energy (LCOE) Calculator Documentation—National Renewable Energy Laboratory (NREL). Available online: https://www.nrel.gov/analysis/tech-lcoe-documentation.html (accessed on 26 February 2021).
- Oliveira-Pinto, S.; Stokkermans, J. Assessment of the potential of different floating solar technologies—Overview and analysis of different case studies. *Energy Convers. Manag.* 2020, 211, 112747. [CrossRef]
- 76. Mostafaeipour, A.; Rezaei, M.; Jahangiri, M.; Qolipour, M. Feasibility Analysis of a New Tree-Shaped Wind Turbine for Urban Application: A Case Study. *Energy Environ.* **2019**, *31*, 1230–1256. [CrossRef]
- Tao, J.Y.; Finenko, A. Moving beyond LCOE: Impact of various financing methods on PV profitability for SIDS. *Energy Policy* 2016, 98, 749–758. [CrossRef]
- Nguyen, S.; Peng, W.; Sokolowski, P.; Alahakoon, D.; Yu, X. Optimizing rooftop photovoltaic distributed generation with battery storage for peer-to-peer energy trading. *Appl. Energy* 2020, 228, 2567–2580. [CrossRef]
- Inns Regulatory Framework-Town of Banff, Canada. Available online: https://banff.ca/DocumentCenter/View/6974/ (accessed on 5 June 2021).
- Ferrari, S.; Zagarella, F.; Caputo, P.; D'Amico, A. Results of a literature review on methods for estimating buildings energy demand at district level. *Energy* 2019, 175, 1130–1137. [CrossRef]
- 81. Wang, J.C. A study on the energy performance of hotel buildings in Taiwan. Energy Build. 2012, 49, 268–275. [CrossRef]
- Lamagna, M.; Nastasi, B.; Groppi, D.; Nezhad, M.M.; Garcia, D.A. Hourly Energy profile determination technique from monthly energy bills. *Build. Simul.* 2020, 13, 1235–1248. [CrossRef]

- Chung, M.; Park, H.C. Comparison of building energy demand for hotels, hospitals, and offices in Korea. *Energy* 2015, 92, 383–393.
 [CrossRef]
- Chen, L.; Lin, S.; Kuo, C. Marketing strategies for the bed and breakfast industry in Taiwan. Int. J. Hosp. Manag. 2013, 32, 278–286. [CrossRef]
- 85. Huang, K.T.; Wang, J.C. Greenhouse gas emissions of tourism-based leisure farms in Taiwan. *Sustainability* **2015**, *7*, 11032–11049. [CrossRef]
- 86. Tsai, W.T. Trend analysis of Taiwan's greenhouse gas emissions from the energy sector and its mitigation strategies and promotion actions. *Atmosphere* **2021**, *12*, 859. [CrossRef]
- 87. Carbon Pricing Dashboard—The World Bank. Available online: https://www.carbonpricingdashboard.worldbank.org/map_data (accessed on 12 April 2021).
- The Price of Electricity Per KWh in 230 Countries. Available online: https://www.cable.co.uk/energy/worldwide-pricing/ (accessed on 16 December 2021).
- Taiwan State Utility Taipower Sees Losses Escalate. Available online: https://www.taiwannews.com.tw/en/news/3746307 (accessed on 16 December 2021).
- Sanyo Hit Power 200 Data Sheet. Available online: https://www.solaronline.com.au/media/attachment/file/p/a/panasonichit-240-v2.pdf (accessed on 5 May 2020).
- Tesla SolarCity SC 320 Data Sheet. Available online: https://sunwatts.com/content/specs/Tesla-SolarCity-SC320-330-96-Cell-Datasheet.pdf (accessed on 23 September 2020).
- 92. Hydrelio Optimized Floating Structure for Solar Plant. Available online: http://www.agse-geologues.fr/documents/2018 /Brochure%20Floating%20Solutions%202018.pdf (accessed on 15 October 2020).
- Catalogue of European Urban Wind Turbine Manufacturers. Available online: http://www.urbanwind.net/pdf/CATALOGUE_ V2.pdf (accessed on 23 September 2020).
- Data Sheet the Wind Tree. Available online: https://reiwind.ie/wp-content/uploads/2021/11/Technical-Data-Sheet-MODULAR-TREE-182430A-Regular-hybrid-models-2021-copie.pdf (accessed on 23 September 2020).
- 95. Reddy, P.; Surendra Gupta, M.V.N.; Nundy, S.; Karthick, A.; Ghosh, A. Status of BIPV and BAPV system for less energy-hungry building in India-a review. *Appl. Sci.* 2020, 10, 2337. [CrossRef]
- 96. Taiwan Power Company Rate Schedules. Available online: https://www.taipower.com.tw/ (accessed on 24 June 2021).